Author’s response to reviews

Title: Aminaphtone's therapy in Type 1 diabetic patients with Albuminuria: A Case Report

Authors:
Concetta Romano (immaromano65@gmail.com)
Consuelo Tamburella (consuelo.tamburella@hotmail.it)
Martino Costa (martinocosta@interfree.it)
Marco Messina (dott.marcomessina@gmail.com)
Anna Lisa Fassari (alfassari@alice.it)
Marco Bertini (bertini@baldaccilab.com)

Version: 2
Date: 4 July 2014

Author’s response to reviews: see over
COVER LETTERS

Pise, July 4th 2014

Dear Editor-in-Chief

I am sending you our manuscript entitled: “AMINAPTHONE’S THERAPY IN TYPE 1 DIABETIC PATIENT WITH ALBUMINURIA: A CASE REPORT” by Concetta Romano, Consuelo Tamburella, Martino Costa, Marco Messina, Anna Lisa Fassari and Marco Bertini. We would like to have the manuscript considered for publication in “JOURNAL OF MEDICAL CASE REPORTS”.

Please find enclosed our manuscript, “AMINAPTHONE’S THERAPY IN TYPE 1 DIABETIC PATIENTS WITH ALBUMINURIA: A CASE REPORT”, which we would like to submit for publication as a “CASE REPORT” in “JOURNAL OF MEDICAL CASE REPORT”.

To our knowledge, this is the first report showing an albuminuria control in type 1 Diabetic Patient by using a vascular drug like AMINAPTHONE. These results significantly advance our understanding in IDDM (Insulin Dependent Diabetes Mellitus) microvascular etiology confirming that a microvascular protector, like AMINAPTHONE, could be useful for control IDDM complications.

We strongly believe that our findings would appeal to the readers of JOURNAL OF MEDICAL CASE REPORTS and that this original case report could be of interest will have a broader clinical impact across more than one area of medicine (Diabetologists, angiologists, nephrologists and general practitioners).

We also confirm that this manuscript has not been published elsewhere and is not under consideration by another journal.

All Authors have approved the manuscript and agree with its submission to “JOURNAL OF MEDICAL CASE REPORTS”.

Please address all correspondence to: Marco Bertini, MD, Medical Department Laboratori Baldacci SpA Via San Michele degli Scalzi 73 – 56124 Pise (Italy) email: bertini@baldaccilab.com

Sincerely yours

Marco Bertini MD
Ph.D in Clinical Pharmacology
Ph.D in Paediatrics